Author's response to reviews

Title: Branchial cysts: an unusual cause of a mediastinal mass: a case report

Authors:

Vihar R Kotecha Dr (viharkotecha@hotmail.com)
Alex Muturi Dr (almuturi@gmail.com)
Josiah Ruturi Dr (ruturi@gmail.com)

Version: 3 Date: 26 March 2015

Author’s response to reviews:

In the abstract the word boy has been replaced with "male" on page 2

Consent section has been changed as per recommendation to read

"Written informed consent was obtained from the patient’s legal guardian(s) for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."